Author's response to reviews

Title: Congenital Arhinia

Authors:

Guzin Akkuzu (guzina@baskent-ank.edu.tr)
Babur Akkuzu (babura@baskent-ank.edu.tr)
Erdinc Aydin (erdinca@baskent-ank.edu.tr)
Murat Derbent (muratd@baskent-ank.edu.tr)
Levent Ozluoglu (levento@baskent-ank.edu.tr)

Version: 2 Date: 17 April 2007

Author's response to reviews:

A competing interests section, an authors' contributions section, an acknowledgements section and a list of abbreviations are added. Spelling of the references section and style of reference numbers are corrected.